Author’s response to reviews

Title: Misleading about MBT in Oslo

Authors:
Sigmund Karterud (wkarteru@online.no)
Elfrida Kvarstein (elfkva@ous-hf.no)

Version: 2 Date: 11 May 2019

Author’s response to reviews:

Dear Samuel Harris,

As I explained to you by email, I am tired of this bureaucracy which steals time and energy from me. The case is very simple. Your journal has conveyed misleading information about our research and it is a professional duty from your side to correct that. The simplest way to do that is to publish our comment. If not, you have to correct it otherwise. If we do not see any appropriate action from your side, the case will be lifted to a higher level by involving Norwegian research authorities.

As you well see I have resubmitted the same text.

Regards,

Sigmund Karterud